Author's response to reviews

Title: A hypothyroid child with rhabdomyolysis and recent history of cardiac tamponade: a case report

Authors:

Juan J Delgado Hurtado (juandelgado@ufm.edu)
Waleska Guevara (waleskaguevara@gmail.com)
Evelyn Ramos (doresly@hotmail.com)
Claudia Lorenzana (claudialorenzana17@yahoo.com)
Susana Soto (susy2874@yahoo.com)

Version: 4 Date: 31 August 2011

Author's response to reviews: see over
Dear editors and reviewers,

Thank you for the comments you made regarding our manuscript. These were valuable and are appreciated.

Following are our responses to each comment or question.

**Referee 1:**
**Comment:**
The manuscript is very interesting. Authors' conclusions about this case report are correct. Indeed a cardiac tamponade correlated with an hypothyroidism is not frequent even in very young patients and this is the real remarkable meaning of the job. Exposition of case report is complete and exhaustive. I hoped to have many more details about the haemodynamic failure during the tamponade (x-rays, ECG, echo, gas analysis) but i must suppose that authors couldn't collect them 2 months after the first admission. This is the only limit of the work. **Quality of written English:** Needs some language corrections before being published.

**Response:** Thank you very much for your comment. Indeed we included the details of the previous pericardial tamponade that were available.

**Referee 2:**
**Comment:**
There are only a few minor grammatical errors that need to be corrected. **Quality of written English:** Needs some language corrections before being published.

**Response:** Thank you very much for your comment. We have corrected the grammatical errors.

Additional changes we have made to the manuscript are: (we think they improve the quality of the case report)

1. For indexing purposes included “Hurtado” as a first authors second last name
2. Improved grammatical errors
3. Extended the description of the figures:
   a. Figure 1: included “…causing right atrial and ventricular collapse”
   b. Figure 2: included “…showing a normal absorbing thyroid”
   c. Figure 3: included “eosinophilic muscle fibers”
4. Included- Figure 4: Muscle fibers with colliquative necrosis

We are open to any other comment or editing suggestion.

Sincerely,
The authors.